

The World's Largest Open Access Agricultural & Applied Economics Digital Library

This document is discoverable and free to researchers across the globe due to the work of AgEcon Search.

Help ensure our sustainability.

Give to AgEcon Search

AgEcon Search http://ageconsearch.umn.edu aesearch@umn.edu

Papers downloaded from **AgEcon Search** may be used for non-commercial purposes and personal study only. No other use, including posting to another Internet site, is permitted without permission from the copyright owner (not AgEcon Search), or as allowed under the provisions of Fair Use, U.S. Copyright Act, Title 17 U.S.C.





Working Paper 021.2020

Investigating the Origins of Differentiated Vulnerabilities to Climate Change and their Effects on Wellbeing

Federica Cappelli

Firms and Cities Towards Sustainability Series Editor: Stefano Pareglio

Investigating the Origins of Differentiated Vulnerabilities to Climate Change and their Effects on Wellbeing

By Federica Cappelli, Roma Tre University

Summary

The paper traces the evolution the concept of socioeconomic vulnerability to climate change has followed in the academic and scientific debate. The recent recognition of vulnerability as a social construction has shifted the focus of the analysis on the dimension of adaptive capacity, i.e. the human response to environmental stresses and climate hazards. Depending on the social and institutional factors in place in a society, this human response can either mitigate or exacerbate harm. In particular, the paper focuses on the role of both vertical and horizontal inequalities in determining differentiated vulnerabilities across individuals and population groups. Inequality in the distribution of income and relevant resources is primarily a question of access to tangible and intangible assets and capabilities that can enable individuals or population groups to prevent and cope with impacts from extreme weather events.

Keywords: Climate Change, Vulnerability, Inequality

JEL Classification: Q54, D63, 015

Address for correspondence: Federica Cappelli Roma Tre University Department of Economics Via Silvio D'Amico, 77 00145 Roma Italy E-mail: Federica.cappelli@uniroma3.it

The opinions expressed in this paper do not necessarily reflect the position of Fondazione Eni Enrico Mattei Corso Magenta, 63, 20123 Milano (I), web site: www.feem.it, e-mail: working.papers@feem.it

Investigating the Origins of Differentiated Vulnerabilities to Climate Change and their Effects on Wellbeing

Federica Cappelli, Roma Tre University

Abstract

The paper traces the evolution the concept of socioeconomic vulnerability to climate change has followed in the academic and scientific debate. The recent recognition of vulnerability as a social construction has shifted the focus of the analysis on the dimension of adaptive capacity, i.e. the human response to environmental stresses and climate hazards. Depending on the social and institutional factors in place in a society, this human response can either mitigate or exacerbate harm. In particular, the paper focuses on the role of both vertical and horizontal inequalities in determining differentiated vulnerabilities across individuals and population groups. Inequality in the distribution of income and relevant resources is primarily a question of access to tangible and intangible assets and capabilities that can enable individuals or population groups to prevent and cope with impacts from extreme weather events.

Keywords: Climate change; Vulnerability; Inequality.

J.E.L. code: Q54; D63; O15

1 Introduction

According to the Global Climate Risk Index (CRI) 2018, developed by the Germanwatch, 524,000 people died across world countries as a direct consequence of more than 11,000 climate-change related extreme events between 1997 and 2016 (Eckstein et al., 2017). The same report shows that global distribution of climate-change related risks and vulnerabilities is largely unfair. Of the ten countries most affected by climate change impacts between 1997 and 2016, nine are categorized as low income or lower-middle income countries. Further, the report considers world countries as belonging to two different groups: on the one hand, countries affected by exceptional catastrophes; on the other hand, countries experiencing continuous threats. The latter group is mainly composed of developing countries (Eckstein et al., 2017).

As extensively reported by Otto et al. (2017), several definitions of socioeconomic vulnerability to environmental stressors have been proposed since the Eighties. A certain degree of uncertainty is due to the extensive use of the concept by different disciplines, all with slightly different acceptations (Füssel, 2012). What pools these definitions is the centrality of the interaction between human systems and the environment. However, in disciplines like climate change research, multidisciplinary by definition, this multitude of definitions may be somewhat problematic (Newell et al., 2005). In addition, the dynamic nature of social vulnerability requires a definition of the concept adaptable to changing institutional contexts (Adger and Kelly, 1999) and environmental and socioeconomic conditions (Cutter and Finch, 2008). The uncertainty posed by climate change about the timing and intensity of its manifestations and the consequent impacts it has on societies and population groups contributes to increasing haziness when trying to define socioeconomic vulnerability.

The present paper reviews the literature on socioeconomic vulnerability to climate change, with a focus on the role of structural inequalities as drivers of such differentiated vulnerability across individuals and population groups. Inequality in the distribution of relevant resources is primarily a question of access to tangible assets, as well as to intangible assets and capabilities that can enable an individual or a population group to prevent and cope with impacts from extreme weather events. In this context, a relevant role is also the one played by institutions and policies, which can either remove obstacles towards a fair distribution of resources or intensify pre-existing inequalities and vulnerabilities. The work is structured as follows: Section 2 reviews the definitions of socioeconomic vulnerability to climate change proposed in the literature and the related approaches. Section 3 addresses the roles of access and governance in shaping vulnerabilities. Section 4 investigates how structural inequalities act as drivers of vulnerability, taking into account both vertical and horizontal inequalities. Section 5 outlines the several threats to wellbeing dimensions posed by increased

vulnerabilities. Section 6 discusses the possible actions to be undertaken to address vulnerability and inequality and concludes.

2 The evolution of the concept of socioeconomic vulnerability to climate change

There have been numerous attempts to provide a reasonable measurement of socioeconomic vulnerability, accounting for the multidimensionality of the concept and comparability across different locations and over time. Some of them measure vulnerability to natural and environmental stressors or exposure to natural hazards at the national level (e.g. International Strategy for Disaster Reduction, 2004; Dilley et al., 2005; Inter-American Development Bank, 2006; Birkmann, 2007). Other studies propose alternative measures of vulnerability strictly dependent on the local background (e.g. Cutter and Finch, 2008) or hazard-specific (e.g. Füssel, 2004; Cannon, 2006; Cardona et al., 2012). Cannon (2006) and Cardona et al. (2012) identify some global trends, including population growth, skewed urbanization, international financial pressures, increasing socioeconomic inequalities, incorrect governance behaviours and environmental degradation, as important drivers of socioeconomic vulnerability.

Further, the concept has evolved from considering a passive (e.g. Tegat et al., 1990; McCarthy et al., 2001) to an active (e.g. Cutter and Finch, 2008; Füssel, 2012; Oppenheimer et al., 2014) role of human systems. Until the first years of 2000s, vulnerability of human systems was only conceived in relation to the physical damages from climate change suffered according to the degree of exposure to adverse events. Hence, differences in the impacts suffered across individuals and population groups were simply explained by the diverse geographical and climatic characteristics of the localities affected. Then, it was observed that in the face of a comparable level of physical impacts, some people suffered harsher harm than others, determining differentiated vulnerabilities across population groups. Hence, the focus has shifted to the ability of human systems to respond to stresses, which can either mitigate or heighten their harm, depending on social and institutional factors. It follows that vulnerability is now conceived primarily as a social construction (Wisner et al., 2004; Thomas et al., 2018), allegedly originating from uneven development processes (Gaillard, 2010).

According to Bohle et al. (1994) social vulnerability to climate change is a multidimensional concept defined by the interaction of three place- and time-specific attributes: i) resource endowments (and their quality); ii) class relations and empowerment, which shape the allocation of the entitlements ($\hat{a} \ la \ Sen^1$) and secure their ownership; iii) political ecology, which depends upon the prevailing theory of political economy and governs the methods of production and the distribution of surplus

¹ Cfr. Sen (1981; 1987; 1990).

across population groups and social classes. In addition to this, several socioeconomic factors define vulnerable populations, including ethnic group, socioeconomic status, age and gender.

Following the same line of evolution, the approach of the Intergovernmental Panel on Climate Change (IPCC) has passed from *climate change adaptation* to *disaster risk management*. In the former (until the report of 2007), vulnerability was understood as a function of exposure, sensitivity and adaptive capacity. It was defined as "the degree to which a system is susceptible to, and unable to cope with, adverse effects of climate change, including climate variability and extremes. Vulnerability is a function of the character, magnitude, and rate of climate change and variation to which a system is exposed, its sensitivity, and its adaptive capacity" [IPCC, 2007 p. 883]. Despite this definition already constituted an advancement compared to the ones provided in previous reports, the focus was still on the exposure to the impacts of climate change. Then, starting from the report of 2014, vulnerability has been understood as a social process, depending only on sensitivity and adaptive capacity. In this framework, the dimension of exposure, although necessary to determining climate impacts, only accounts for the geographical aspects but does not contribute to the creation of vulnerabilities, which have now a purely social and human connotation. With this change of perspective, the process of adaptation to climate impacts has changed accordingly: the adoption of the disaster risk management approach has allowed a shift in the focus from disaster recovery to disaster prevention (Cardona et al., 2003; Cardona et al., 2012). This can only be possible through a deep understanding of the interactions existing between human systems and the environment, as well as of the people and assets exposed to environmental hazards and the social, political and economic factors that determine the distribution of key resources and shape vulnerabilities. In the remainder of the paper, we will draw on the definition proposed by the IPCC in the Fifth Assessment Report of 2014, which depicts socioeconomic vulnerability as "the propensity or predisposition to be adversely affected. Vulnerability encompasses a variety of concepts and elements including sensitivity or susceptibility to harm and lack of capacity to adapt. A broad set of factors such as wealth, social status, and gender determine vulnerability and exposure to climate-related risks" [Oppenheimer et al., 2014; p. 1048].

The definition of exposure has changed together with the approach adopted by the IPCC. Until 2007, it was defined as "the nature and degree to which a system is exposed to significant climatic variations". In the report of 2014, the element of exposure loses its role as driver for vulnerability and assumes a spatial connotation (Jurgilevich et al., 2017). In this latest report, it is defined as "the presence of people, livelihoods, species or ecosystems, environmental functions, services, and resources, infrastructure, or economic, social, or cultural assets in places and settings that could be adversely affected." Accordingly, if a hazard occurs in an area unpopulated and without valuable

resources, technically no disaster can exist, and no dimension of wellbeing can be threatened. For a disaster to exist, a human system needs to be both exposed and vulnerable (Cardona et al., 2012).

The two intrinsic dimensions of vulnerability, as in the report of 2014, are sensitivity and adaptive capacity. In this case, their definitions have remained substantially unchanged between the report of 2007 and the one of 2014. Sensitivity is defined as "the degree to which a system or species is affected, either adversely or beneficially, by climate variability or change. The effect may be direct (e.g., a change in crop yield in response to a change in the mean, range, or variability of temperature) or indirect (e.g., damages caused by an increase in the frequency of coastal flooding due to sea level rise)" [IPCC, 2014]. High sensitivity of a human system is the result of context-specific and intrinsic conditions of the system itself. In other words, sensitivity accounts for the weaknesses that make the system predisposed to suffer damages from the occurrence of a hazard. However, such a damage may be prevented or at least mitigated if the system is endowed with adaptive capacity.

Adaptive capacity is defined as "the ability of systems, institutions, humans and other organisms to adjust to potential damage, to take advantage of opportunities, or to respond to consequences" [IPCC, 2014]. It is a critical property of the system that consists of a set of system-specific attributes (for instance, in an agricultural system, the availability of irrigation systems, as well as the alternance of crops to facilitate terrain regeneration, may be some of these attributes) that enable the system to reduce and respond to the adverse impacts caused by the incidence of a hazard. On the contrary, vulnerability increases either in systems characterized by a lack of adaptive capacity or in consequence of 'maladaptation', i.e. the adoption of actions oriented at reducing impacts from climate change, but which instead "increase the vulnerability of other systems, sectors or social groups" (Barnett and O'Neill, 2010). For example, the construction of large dams aimed at improving water security and increasing the share of energy produced by renewable resources may have as a side effect the dislocation of local populations to more exposed locations, as well as an increase in methane emissions (Kallis, 2008). Further, while effectively preventing the risk of smaller floods, large dams often tend to increase the risk of larger floods if not specifically designed for this kind of eventuality or if inadequately maintained (Etkin, 1999). Hence, floodplains that first were uninhabitable become now home to populations whose vulnerability to large floods is increased, with the result of "risk transference" into the future (Fordham, 1999).

Adaptive capacity is a key element in determining an individual or a community's vulnerability both in prevention of a hazard and when recovering from its occurrence. However, the capacity needs to anticipate (ex-ante) disaster risk differ from those to recover from (ex-post) disaster impacts (Cooper et al., 2008). To foster this kind of capacity both structural and policy interventions are needed. Structural interventions include the construction of hazard-resilient buildings as well as

hazard-preventing infrastructure. Policy interventions include both land and environmental policy prescriptions (e.g. the prohibition to build in unsafe areas) and others that are strictly related to aspects of political economy aimed at securing livelihoods (Eriksen and Silva, 2009), guaranteeing access to and a fair distribution of relevant resources. The capacity to respond to climate impacts is useful both in the aftermath of an extreme event and as a first anticipatory measure in the face of future events (Cardona et al., 2012). In the wake of an extreme event, governments should be able to guarantee adequate care and assistance to affected people as well as a fast and effective reconstruction of the buildings and infrastructure hit. Both ex-ante and ex-post adaptive capacity require a long-term perspective as well as the evaluation of costs and benefits of the interventions to recover from current damages and minimize future climate impacts for all population groups. As argued by Sovacool et al. (2018, p. 243), "sometimes, the human response to a natural disaster can exacerbate its impact, even more than the event itself".

While human societies can be vulnerable to specific climate hazards and not to others (hazardspecific vulnerability), some factors heighten vulnerability independently on the type of hazard (Cardona et al., 2012). These factors, including poverty and inequality, have a socio-political and cultural nature and are what transforms a hazard into a disaster (e.g. Lavell, 2003; Cannon, 2006; Gaillard et al., 2014; Thomas et al., 2018), threatening several dimensions of wellbeing. The sociopolitical and cultural factors are the main focus of an emerging strand of research aimed at showing how climate-induced disasters are in fact "unnatural" and the entity of their damages as well as the timing and modalities of recovery mainly depend on the political economy of the country or region where the hazard has occurred (e.g. Weber and Messias, 2012; Cretney, 2017; Sovacool, 2017; Sovacool et al., 2018). An example is Hurricane Katrina, which hit the coast of Lousiana in 2005 causing the death of 1,800 people (and many others missing and severely injured). Although other cities of Louisiana have been affected more severely, the majority of the victims was concentrated in the city of New Orleans, because of a failure in the levee system meant to protect the city. In consequence of this episode, Hurricane Katrina was defined by a Professor of Civil and Environmental Engineering as "the worst civil engineering disaster" in the history of the USA (Seed, 2007).

Figure 1 represents the logical steps that lead climate change to impact human systems through socioeconomic vulnerability and its dimensions.



Figure 1: Dynamics guiding socioeconomic vulnerability to climate change

Source: Own elaboration

Climate change manifests itself through hazards of diverse typologies, whose impacts affect socioeconomic vulnerability of the human systems exposed. Such vulnerability is commensurate with the sensitivity of the affected system and can be mitigated if the system is endowed with (and is capable of handling adequately) adaptive capacity. Both sensitivity and the presence and effectiveness of adaptive capacity of a system depend strictly on the structural inequalities existing in the system itself. The degree of exposure and vulnerability to climate change impacts may determine various threats to wellbeing dimensions, including food insecurity, health issues, armed conflicts and migrations.

3 Access and governance

As discussed above, individuals and groups are at risk of suffering climate impacts not only because of their exposure to a hazard, but also because of differentiated vulnerabilities, which are the direct consequence of marginalization and lack of access to crucial resources (Watts and Bohle, 1993). Norms and cultural values, human behaviours and governance structures that characterize a society impose socioeconomic and political constraints and shape the allocation of resources in a way that creates vulnerabilities (Wisner et al., 2004). Zamani (2006) observes that vulnerability is proved to intensify in periods of significant socioeconomic transition and political disorders, characterized by unequal access across population groups and social classes to crucial resources.

As suggested by Bohle et al. (1994), the inspiring work by Amartya Sen on famines has many common points with socioeconomic vulnerability to climate change and, in particular, with the dimension of adaptive capacity. In both cases, access to relevant resources is not guaranteed by their

availability, but rather depends on the distribution of entitlements and the role of institutions in defining their allocation. For instance, Hickel (2017) highlights the issue of access through the examples of the massive exportation of food from Ireland during the potato famine of 1845-49, as well as from India during the numerous famines occurred between 1850 and 1899, despite the starvation that was affecting the population. Schröter et al. (2005) adopt just the example of famines to describe the social dimension shaping vulnerability to climate change. They suggest focusing on the marginalization of individuals as induced by social, political and institutional factors as the main cause of vulnerability, rather than on the physical impacts the individuals are subject to. Similarly, Adger and Kelly (1999) extend Sen's (1990) analysis of food insecurity to provide an assessment of climate vulnerability that grounds on the 'architecture of entitlements'. The authors maintain that the human use of resources, which in its material and social aspects is regulated by the entitlements of individuals and groups, is key in understanding differential vulnerabilities to climate impacts.

Relevant resources in the face of climate change include both private – e.g. private capital, insurance, owned assets - and public goods – e.g. disaster warning systems, emergency response systems, infrastructure, information and communication technology, etc. (e.g. Thomas et al., 2018; Cinner et al., 2018). The lack of access to these resources is what jeopardizes a system's adaptive capacity to prevent and recover from climate hazards. Vulnerability has both a subjective (endogenous) and a contextual (exogenous) dimension (Lewis, 1999; Wisner et al., 2004; Bankoff et al., 2004; Hore et al., 2018). The exogenous dimension comprehends all socioeconomic, political and institutional factors determining access to relevant resources for individuals and groups. The endogenous dimension is related to the inability of people and communities to secure their access to resources, as a result of unawareness and lack of livelihood choices (Gaillard, 2010; Hore et al., 2018). However, this endogenous dimension is largely influenced by exogenous factors as well (Hore et al., 2018).

The primary origin of differentiated access to resources among individuals and groups are the power relations in place in a society. According to these relations some individuals and groups are included within political decisions, while others are left out, marginalized. Several ways in which marginalization takes place exist: according to Gaillard (2010, p.222) "disaster-affected people are marginalised geographically because they live in hazardous places (e.g. informal settlers); socially because they are members of minority groups (e.g. ethnic or caste minorities, disabled individuals, prisoners and refugees); economically because they are poor (e.g. homeless and jobless); and politically because their voice is disregarded (e.g. women, non-heterosexuals, children, and elderly) by those with political power." Adger and Kelly (1999) distinguish between the factors determining individual and collective vulnerability. They maintain that while the former depends on the

individual's social status within the community of reference, access to resources and the possibility of diversification of income sources, the latter is related to wider characteristics of society, including institutional and market structures, infrastructure and income, which are all worsened by climate change.

While the theoretical paradigm shift that has allowed to consider climate-related risks and disasters as social constructions, as opposed to natural phenomena, policy prescriptions aimed at minimizing and coping with such risks have not followed the same direction, yet (Gaillard, 2010; Hore et al., 2018). Albeit there are several examples of local institutions that managed to handle scarce relevant resources (Kallis and Zografos, 2013), emphasis is still placed on technological and engineering solutions aimed at preventing the physical impacts of a specific array of extreme climate events, often forgetting about the vulnerability of local populations (Gaillard, 2010). As noted by Gilbert (1995), these interventions typically resemble war strategies and are often handled by the army. This is not to say that technological solutions are not effective or important in preventing these kinds of hazards but that, in accordance with the recognition of the social essence of vulnerability, should be implemented in tandem with social and political solutions (Kelman et al., 2015). Sovacool (2018) contends that socioeconomic vulnerability should be evaluated at the local level, where four mechanisms inducing increased vulnerabilities can take place, namely enclosure, exclusion, encroachment and entrenchment. By enclosure, Sovacool (2018, p. 186) refers to "when an adaptation project transfers a public or social asset into private hands, or expands the role and authority of a private actor into a formerly public sphere". Exclusion occurs "when an adaptation project excludes or displaces a particular group of stakeholders or limits access to resources related to due process, fairness, and procedural justice" (ibid). Encroachment takes place "when adaptation projects degrade the environment, interfere with ecosystem services provision, intrude upon biodiversity conservation zones such as protected areas and national parks, or counteract climate change mitigation efforts by involving the emission (embodied, or direct) of greenhouse gases" (ibid). Finally, entrenchment is "when an adaptation project aggravates vulnerability or the disempowerment of women, the chronically poor, and/or other minority groups" (ibid).

For instance, D'Alisa and Kallis (2016) report the example of maladaptation that had place in Sarno (Italy), after the occurrence of the landslide that affected the area in 1988. According to the authors, maladaptation was the result of the choice to implement harder engineering interventions over softer interventions. By "hard interventions" the authors refer to large-scale structural engineering interventions, like dams, canals, etc; by "soft interventions" they refer to structural ecological interventions aimed at restabilising natural functions of the physical environment as well as non-structural interventions at the institutional level. Harder interventions are usually preferred by

public institutions over softer interventions because of the huge amounts of capital they help circulate but, especially if implemented alone, are rarely more effective and are often source of increased vulnerability of the areas concerned (D'Alisa and Kallis, 2016). Further, hard interventions not accompanied by soft interventions can often have the undesired effect of risk transference to other vulnerable groups (Hore et al., 2018).

At the international level, this trend reflects the idea of widely applicable development policies whereby rich and safe countries are opposed to poor and endangered countries (Gaillard, 2010). The result is a transfer of knowledge and technologies from developed to developing countries, that once again does not account for local-specific differentiated vulnerabilities. Conversely, when assessing vulnerability of individuals or population groups the local socio-economic and political context where adaptation takes place is of extreme importance. Several studies have documented the local power implications of adaptation plans in different countries. For instance, in Kenya some adaptation programs have altered power relationships so as to reinforce the relative power of some pastoralist communities who, thanks to their gained powerful status, managed to exclude other pastoralist groups from the access to the main water basin of the region (Eriksen and Lind, 2009). Analogously, other adaptation projects in Ghana resulted in a disproportionate workload between men and women, with women more heavily burdened (Carr, 2008). Sovacool (2018) investigates the consequences of the implementation of the National Adaptation Program of Action in Bangladesh and finds that the processes of enclosure, exclusion, encroachment and entrenchment have hindered an equitable adaptation, favouring some local elites to the detriment of other population groups. These examples are to remind that, as emphasised by Gaillard (2010), while no risk reduction policy can be conceived to fit everywhere, fostering participation of local communities and accounting for local power relations is a prerequisite for reducing vulnerability.

4 Structural inequalities as drivers of socioeconomic vulnerability

Barrett (2013) argues that climate change involves "a double inequality", determined by the opposite distribution of impacts and responsibilities. As also discussed during the First Conference of Parties (COP-1) of the UNFCCC, held in Berlin in 1995, while developed countries have historically been the major contributors to climate change impacts through their industrialization process, developing countries are the most affected, without even benefitting from economic growth. Further, the disproportionate impacts they suffer contributes to widening pre-existing inequalities (IPCC, 2001) and exacerbating global poverty (Harris, 2013). Most studies on the interaction between climate impacts and inequality have focused on inequalities between regions. However, in this way

climate-change induced poverty and inequality within regions are hidden behind regional averages (Barrett, 2013).

Further, such inequality is not a feature characterizing only developing countries. It can be found across and within regions and cities of the wealthiest countries as well. For instance, the unequal distribution of impacts from climate change is well represented in figure 2, sourced from the study by Hsiang et al. (2017), which represents the projection of the spatial distribution of climate-change induced damages in the USA counties as a percentage of the counties' GDP.





Source: Hsiang et al. (2017)

In their study, Hsiang et al. (2017) develop a probabilistic model of future climate impacts at a spatially disaggregated level for the USA counties under a RCP8.5 scenario. They first present results at the sectoral level and then aggregate them to obtain a single measure of overall damage. The authors show that impacts from climate change are projected to be extremely differentiated across counties, with the South-Eastern ones especially affected. Further, while median losses are expected to exceed 20% of GDP in some counties, in other counties, mainly located in the North-West, median gains are expected to exceed 10% of GDP. Figure 2 demonstrates that even in wealthier countries climate change tends to exacerbate pre-existing inequalities (Hsiang et al., 2017). Indeed, the distribution of damages reflects the current distribution of income across USA counties, with richer counties being less affected and poorer counties most heavily impacted.

Further, the concept of environmental justice was born in the United States of America during the contestations of the "Warren episode". In 1982, the USA government was to decide where to dispose of the highly toxic wastes illegally dumped across 14 counties in North Carolina (Banzhaf et al., 2019). The ultimate choice fell on the Warren County, where 60 percent of the population was

African-American and 25 percent of families were below the poverty line. The wave of protests in consequence of this decision gave birth to the environmental justice movement and to the acknowledgement of the issues related to environmental injustices by governmental and academic institutions. The US Environmental Protection Agency (EPA) defines environmental justice as "the fair treatment and meaningful involvement of all people regardless of race, color, national origin, or income with respect to the development, implementation and enforcement of environmental laws, regulations and policies". Nonetheless, despite the institutional acknowledgement and the visibility within the public debate, as well as the institution of the "Environmental Equity Workgroup" by the EPA, the USA are still far from achieving environmental justice (Laurent, 2011). Similarly to the Warren episode, several studies (for example, Pfeffer, 2002; McKenzie et al., 2016; Ash and Boyce, 2018) have shown how landfills and other pollution sources have been regularly located in areas characterized by ethnic diversity or social and economic deprivation.

Following the diffusion of the concept of environmental justice, the notion of climate justice has gotten a foothold among researchers and environmentalist movements, with the specific aim of wondering about the uneven distribution of risks and responsibilities across countries and population groups (Barrett, 2013).

The presence of widespread inequalities may intensify exposure and vulnerability of specific individuals and groups by inducing both an increase in the sensitivity to climate hazards and a lack of adaptive capacity to prevent and recover from the occurrence of extreme events. In general, differentiated vulnerabilities can be seen when, for the same level of exposure, some individuals or groups are more vulnerable than others (Birkmann, 2007). Higher sensitivity is often induced by the limited ability of poor people to choose where to live, being forced to opt for cheaper but often less resistant buildings (Hallegatte, 2012), as well as localities exposed to hazards (Heijmans, 2001; Yamamura, 2015). Exposure and sensitivity to climate impacts result from poverty and underdevelopment (Wisner et al., 2004) and the adverse consequences are amplified by poor-quality health and education infrastructures and institutions (Mearns and Norton, 2009). Trade governed by market forces and poor regulations does not take into account issues of justice and can have the unintended effect of enhancing inequalities in the distribution of impacts. The increase of food and resource prices induced by trade and finance, if not counterbalanced by distributive and supportive policies, raises vulnerability of households in the lowest tails of income distribution (Oppenheimer et al., 2014).

Further, Thomas et al. (2018) distinguish among three types of adaptive capacities necessary to reduce vulnerability: economic, institutional and political capacity. By lack of economic capacity, they refer to limitations in access to resources imposed by income or wealth constraint. By lack of

institutional capacity in a certain group they indicate the inability of that group to gain institutional inclusiveness and consideration in the allocation and provision of relevant public resources. Finally, by lack of political capacity they imply neglected access to key resources as resulting from surplus appropriation by other groups. Despite the lack of either capacity results in the group being poor, policies to be adopted to reduce the group's vulnerability differ depending on the type(s) of capacity lacking (Thomas et al., 2018). Inequalities in access to resources and in social hierarchies that strengthen it can rarely be reverted when adapting to climate impacts. On the contrary, the occurrence of climatic changes and hazards usually tends to reinforce the pre-existing inequalities (Cannon, 1994).

While most discussions about inequality usually focus on vertical inequalities (i.e. income or wealth inequalities between individuals, typically measured by the Lorenz curve and the Gini index), horizontal inequalities (i.e. inequalities between groups of individuals) tend to be disregarded. The acknowledgement of horizontal inequalities as well as their relations with vertical inequalities is especially relevant to understand socioeconomic vulnerability to climate change and issues related to environmental and climate justice. While horizontal inequalities constitute a part of vertical inequalities, which are composed of between and within group inequalities, it is sometimes very difficult to achieve targets such as poverty abatement or vulnerability reduction without explicitly addressing horizontal inequalities and the resulting marginalization of specific population groups (Stewart et al., 2005).

4.1 Vertical inequalities

The concentration of income and wealth in fewer hands constraints coping capacity and limits the possible strategies to recover from environmental stresses (Adger and Kelly, 1999). Poorer households often rely on public coping strategies only, but external aid or the transfer of financial resources that do not explicitly account for climate justice, are not always effective in reducing vulnerabilities of low-income households (Scandrett, 2016).

Wilkinson and Pickett (2010) start from the premise that the long-time trend of rich societies of valuing economic growth above income equality, has produced manifold negative outcomes in terms of wellbeing. The authors show, by means of a series of scatterplots that relate income inequality with several health, social and environmental indicators, that the unequal distribution of income strongly affects the quality of life even in richer nations. As they argue, "it often looks as if the effect of higher incomes and living standards is to lift people out of these problems. However, when we make comparisons between different societies, we find that these social problems have little or no relation to levels of average incomes in a society" [Wilkinson and Pickett, 2010; p. 11]. They take health as

an example, considering only rich countries. They show that while a country can be twice as rich as another, it does not translate in any improvement in terms of life expectancy. Then, focusing on the USA only, they show that death rates are systematically associated with the level of income, with people in the lowest tails of the distribution being always more affected.

Wilkinson and Pickett (2010) contend that income inequality is representative of the degree of a society's "social stratification". In countries characterized by a high social stratification, there are usually also higher fragmentation, hierarchy, and minimal interactions between different social groups. According to the authors, material inequalities lay the foundations for the formation of class and cultural differences: mere differences in income and wealth evolve over time in differences in attributes shaping class identity, including differences in clothing, consumption, education possibilities, etc.

Social stratification and the differentiated possibilities in consumption it entails is what also determines vulnerability, as resulting from differences in the affordability of mechanisms preventing from and coping with climate-change induced hazards. Explicitly addressing differentiated vulnerabilities when designing preventing and recovery strategies from extreme events is essential to avoid widening the inequality gap and exacerbating pre-existing vulnerabilities. Sovacool et al. (2018) report as an example the severe earthquake that struck Canterbury, New Zealand, in 2010. In that occasion, a fifth of the population affected was not covered by insurance and the recovery process was entrusted to market mechanisms that did not account for vulnerabilities and income differentials (Sovacool et al., 2018). As a result, the authors report that most resources were devoted to the reconstruction of the economic activities located in the city centre, leaving poorer households behind and widening the inequality gap.

4.2 Horizontal inequalities

Categorisation in groups can take place based on different aspects (e.g. ethnic group, gender, age, class, caste, etc.) and can be the result of self-identification, identification by others or a combination of both (Stewart et al., 2005; Stewart, 2009). Being part of a group can sometimes influence positively or negatively the wellbeing of individuals belonging to it, especially when: i) individual's mobility across groups is impossible or very difficult; ii) individuals are treated differently (privileged or discriminated) because belonging to a certain group; iii) a relevant part of an individual's identity is related to group membership (Stewart et al., 2005). In these cases, the individual's wellbeing does not depend solely on their own social status and access to resources, but also on their group's power relationship compared to other groups.

Several studies have investigated differentiated vulnerabilities to climate-related hazards, as resulting from ethnicity (e.g. Peacock, 1997; Fothergill et al., 1999; Elliott and Pais, 2006; Cutter and Finch, 2008; Bonilla-Silva, 2013; Hardy et al., 2017; Sovacool et al., 2018), socioeconomic status and class (e.g. Sen, 1981; Schmidt, 2005; Ray-Bennett, 2009; Arora-Jonsson, 2011; Sovacool et al., 2018), gender (e.g. Peacock, 1997; Enarson and Morrow, 1998; Denton, 2002; Ray-Bennett, 2009; Houghton, 2009; Sultana, 2010; 2014; Arora-Jonsson, 2011; Sovacool et al., 2018), and age (e.g. Jabry, 2003; Stafoggia et al., 2006; Bartlett, 2008; Gosling et al., 2009; Kunz et al., 2013).

Ethnic groups

Structural racism, which comprehends "the totality of the social relations and practices that reinforce white privilege" (Bonilla-Silva, 2013, p. 9), is not as evident as open acts of racism were in the past but, analogously, contributes to the preservation of white privileges through creating less visible inequalities in access to resources (Shearer, 2012). This results in disparities in educational attainment, income and wealth (Thomas et al., 2018) that constrain adaptation possibilities to climate change and increase vulnerability of marginalized ethnic groups. Sovacool et al. (2018) analyse some of the major climate-related disasters occurred in the last years to detect the presence of the mechanisms of enclosure, exclusion, encroachment and entrenchment. In particular, marginalization relates to the mechanism of entrenchment, according to which some ethnic groups may be located in areas where sensitivity and exposure are higher or may be discriminated in the phase of disaster recovery. For instance, the authors show how the mechanism of entrenchment took place when Hurricane Katrina hit the coast of Louisiana, USA. African Americans underwent both disproportionate impacts (Driesen et al., 2005) and uneven recovery processes (Masozera et al., 2007; Bullard and Wright, 2009) in the aftermath of the hurricane, compared to white people. Analogously, the entrenchment mechanism was found in the disproportionate disbursement of official aid across religious groups in Thailand after the tsunami that hit Malaysia, Thailand, India, East Africa and Sri Lanka in 2004 (Sovacool et al., 2018).

Age groups

Children and older people are especially vulnerable to climate-related extreme events. Bartlett (2008) argues that children's incomplete physical development as well as their immature immune system make them more susceptible to vector-borne diseases. Further, both children and older people are more vulnerable to heat-related stress because of their reduced thermoregulatory capacity (Grundy, 2006; Hajat et al., 2007). Children and the elderly are also more vulnerable in the aftermath of the event, as they lack capacity to cope with hazards. For instance, when Superstorm Sandy that

hit New York and New Jersey in 2012, most deaths involved older people because of their constrained access to healthcare and transportation (Kunz et al., 2013).

Gender

Contrary to vulnerabilities related to age structures, gender-related vulnerabilities do not depend on biological differences between men and women. In most cases, they are driven by socio-political and institutional factors that cause women and girls to be relegated to working in jobs more exposed to hazards, or to be excluded from decision-making processes (Al Nahian et al., 2013) and access to relevant resources (Denton, 2002; Sultana, 2014) or, again, to be discriminated during the phase of recovery from a hazard (Houghton, 2009; Sultana, 2010; Al Nahian et al., 2013; Sovacool et al., 2018). For instance, in relation to the latter aspect, Sovacool et al. (2018) report of a gender and rural bias in the disbursement of aid in the Philippines, in the aftermath of Typhoon Yolanda in 2013. Further, in some countries gender-related vulnerabilities are also related to religious, cultural and class factors (Ray-Bennett, 2009; Arora-Jonsson, 2011), which may hinder women's possibilities to evacuate and recover from extreme events in the absence of the male head of the family (Sultana, 2014).

Social status

In several countries, discrimination takes the form of hierarchical relations among individuals belonging to different castes. Such castes can be either officially or not officially acknowledged, but in both cases the type of relationship that is established can deeply influence differentiated vulnerabilities to climate change impacts (Ray-Bennett, 2009). The unequal social status characterizing castes leads to the establishment of patron-client relationships between people belonging to different castes, often bringing about exploitation processes and dependencies of people of lower castes (Onta and Resurreccion, 2011). Such dependency shapes access to food, water, land, education, disaster warning systems and recovery mechanisms (Mustafa et al., 2010; Ray-Bennett, 2009) and, in countries like Nepal, it often determines political exclusion of people in lower castes (Nagoda and Nightingale, 2017).

5 Vulnerability-driven threats to wellbeing dimensions

Several studies (e.g. Oppenheimer et al., 2014; Otto et al., 2017) identify four main dimensions of wellbeing that are threatened by increased socioeconomic vulnerability to climate change: i) food security; ii) health; iii) safety; iv) migrations and displacements. The related threats will be examined in the following paragraphs. It is worth noting that multiple threats may occur at the same time and

place, both independently one from another and as a result of the interaction among different coexisting threats. The same individual or population group can be vulnerable to more than one wellbeing dimension and, once a dimension is affected, their own vulnerability to other threats is likely to increase.

Food insecurity

As observed in the previous sections in relation to famines, socioeconomic vulnerability to climate change and food insecurity share several aspects related to climate hazards. Food security is defined by the Food and Agriculture Organization (1996) as composed of four dimensions: "(i) the availability of sufficient quantities of food of appropriate quality, supplied through domestic production or imports (including food aid); (ii) access by individuals to adequate resources (entitlements) for acquiring appropriate foods for a nutritious diet; (iii) utilization of food through adequate diet, clean water, sanitation, and health care to reach a state of nutritional well-being where all physiological needs are met; and (iv) stability, because to be food secure, a population, household or individual must have access to adequate food at all times. They should not risk losing access to food as a consequence of sudden shocks (e.g. an economic or climatic crisis) or cyclical events (e.g. seasonal food insecurity). The concept of stability can therefore refer to both the availability and access dimensions of food security."

Climate-induced food insecurity can be due to either decreasing crop yields (Das Gupta, 2013) or increasing food prices (Nelson et al., 2009). Several impacts on climate-induced food security are already appreciable from observed data (Lobell et al., 2011) and future projections are anything but reassuring. The Fifth Assessment Report (AR5) of the IPCC confirms, supported by new evidence, conclusions to which AR4 came to on climate change-induced impacts on food security (Oppenheimer et al., 2014). According to the study, climate change will have differentiated impacts on food security in terms of both latitudes where vulnerable populations and food productions are located and yields of different crop typologies. Low-latitude and dryland areas are generally found to be the most at risk (Das Gupta, 2013; Oppenheimer et al., 2014). In these regions, even small increases in local temperatures can be detrimental to crop yields and threaten food security of local populations. Maize and wheat yields are found to be affected already with less than 2°C local temperature increase in vulnerable areas, while temperate maize and tropical rice will be affected in consequence of a 3 to 5°C local temperature increase (Oppenheimer et al., 2014).

Simulton et al. (2012) reach similar conclusions about crops vulnerability and contend that the same crops respond differently to droughts if produced in developed (where investments in agriculture are higher) or developing countries. Kotir (2010) reviews several works trying to predict

future food security, agricultural potential and crop productivity in Sub-Saharan Africa under different climate change scenarios and employing different methodologies (namely, crop simulation models, agro-ecological zones models, cross-sectional Ricardian models). He reports that, despite considerable variations in the magnitude of impacts found by different studies and methodologies, all examined works agree on the future climate-induced decrease in the area suitable for agriculture, the length of growing seasons and crop yield potential, especially in semi-arid and arid areas of Sub-Saharan Africa. Several other case studies have investigated the differentiated impacts of climate change on food security in many countries (e.g. Amarasinghe et al., 2005; Challinor et al., 2007; Ahmad et al., 2011; Shepherd et al., 2013; Birthal et al., 2014; Boko et al., 2018). Further, several studies agree on projections that see climate change as severely exacerbating both water scarcity and the frequency of floods in the next decades (e.g. Schewe et al., 2013; Gosling and Arnell, 2013). Limited availability and access to freshwater resources can seriously threaten a fair utilization of and access to food resources as well (Wheeler and von Braun, 2013).

As for the access dimension, an increase in food prices may seriously undermine poorer households' access to food, compromising their food security. In addition, following Engel's law, poorer households tend to spend a higher share of their income on food and essential resources compared to richer households, hence an increase in food prices likely contributes to widening the inequality gap. Among strategies to cope with food insecurity, Otto et al. (2017) report the switch to cheaper and lower-quality staples, the reduction of quantities of food ingested (often by women who sacrifice themselves to leave more food for children and husband), the placement in the job market of other family members.

Health issues

A 2009 The Lancet article declares that "climate change is the biggest global health threat of the 21st century" (Costello et al., 2009). Some recognized health effects from climate change include vector-borne diseases, water-borne diseases, food-borne illnesses and malnutrition, heat-related morbidity and mortality, injuries from extreme weather events, cardiovascular and lung diseases from air pollution, increased aeroallergens from poor air quality. Other health effects are likely to result from climate change but not openly acknowledged. Chapter 11 of the AR5 by the IPCC (Smith et al., 2014) is all devoted to the assessment of health effects as resulting from climate change. In particular, they review the English-language literature on climate-induced health impacts up to August 2013. Their main findings suggest that climate change has both direct and indirect effects on human health, despite the presence of several gaps in the literature, especially for what concerns studies focusing on developing countries and highly vulnerable populations groups. Direct effects include health impacts

directly related to changes in temperatures, precipitations and drought levels (Smith et al., 2014). In particular, heat-related impacts are especially relevant, especially for younger and older people (Grundy, 2006; Hajat et al., 2007). Further, heat-related impacts also reflect differentiated vulnerabilities, as disadvantaged households may not afford air-cooling or may be more exposed because of the job in which they are employed (Smith et al., 2014) or because they are homeless (Walters and Gaillard, 2014). Results on the association between increasing number of hot days and related deaths are rather robust (Honda et al., 2014), as opposed to those related to other climatic phenomena which are more inconclusive. The IPCC (2012) has shown that, on a global scale, the total number of cold days and nights has decreased, against an increase in the total number of hot days are found in the case of floods (e.g. Pradhan et al., 2007; Khan et al., 2011).

Indirect effects are mediated by either natural systems (e.g. air pollution, water-borne diseases) or human systems (e.g. undernutrition, mental health) or a combination of the two (Smith et al., 2014). Even in the case of climate-related health issues, inequality in access to healthcare systems acts as a driver of differentiated vulnerabilities among individuals and groups. Several studies (e.g. Kallis, 2008; Stanke et al., 2013; Sena et al., 2014; Yusa et al., 2015; Ebi and Bowen, 2016) have documented the (mainly indirect) health effects of long-term droughts. All studies agree on the fact that low-income countries are the most heavily affected and that health-related impacts of droughts are the result of socioeconomic issues, including poverty, food and water insecurity and displacement. Both direct and indirect health effects, as well as the prevalence of some infectious diseases over others are strictly local-specific, depending on a multitude of interacting socioeconomic, political and environmental factors (World Bank, 2010). Further, future impacts of climate change are projected to aggravate existing illnesses in places where are already widespread, as well as to extend them to other localities and to favour the emergence of new illnesses (IPCC, 2014).

Conflicts

Armed conflicts are a serious threat to livelihoods and development, so much that Collier (2003) define them as "development in reverse", because of the costs they impose to societies and the lingering effects they have on the whole economy for several years after the conflict has ended. The nature of armed conflicts has changed substantially since the foundation of the United Nations 75 years ago, becoming on average less lethal (United Nations, 2020); despite that the number of conflicts has risen over this period. The year 2016 registered the highest number of conflicts over the last 30 years (United Nations and World Bank, 2018) and it is estimated that, by 2030, more than half

of the world's population living in developing countries will be afflicted by acute violent episodes (OECD, 2015).

Collier (1999) argues that the disruptive effects of conflicts on the economies manifest themselves through several mechanisms, including the destruction of resources, social disorder, the shifting of public expenditure towards military purposes and capital flight. Several other studies (Gates et al., 2012; Fukuda-Parr et al., 2011; Shemyakina, 2011; Cohen and Pinstrup-Andersen, 1999; Teodosijevic, 2003) report their effects in terms of exacerbation of food insecurity and development issues. Among these, results from an empirical work by Messer and Cohen (2004) show that civil conflicts in Africa from 1960 to 2000 caused losses amounting to \$120 billion in agricultural production.

In the last decades, the disruptive effects of climate change have been manifesting at a far hastier pace and with a stronger intensity (Hoegh-Guldberg et al., 2018). In the same way as armed conflicts, climate change is a major security concern and affects crucial aspects of human life especially in developing countries, through its impacts on agriculture and food security, access to water and other resources (Markandya et al., 2017). This may exacerbate the social disorder and instability already present in those countries, likely fuelling conflicts (Weir and Virani, 2011). Indeed, even though socio-economic and institutional circumstances are crucial in explaining the onset and evolution of conflicts (Buhaug, 2010), climate change and resources endowment may reasonably act as "threat multipliers". Doherty and Clayton (2011) find that, as a consequence of a 1.1°C (2 degrees Fahrenheit) temperature increase in the USA, interpersonal violence and murders per year would rise by 24,000. In a study of 2013, Hsiang, Burke and Miguel estimate the impact of a one standard deviation increase in either temperature or extreme rainfall on different typologies of conflicts. They find that deviations from normal values have a systematic effect on increasing conflict risk. In particular, such a 1-SD increase brings about an intensification of the frequency of interpersonal violence by 4% and of intergroup conflict by 14%. Furthermore, numerous studies (Gleick, 2014; King, 2015; Kelley et al., 2015) have identified the hard drought conditions affecting Syria during the last years as one of the factors contributing directly to the deterioration of the country's economic conditions and, accordingly, to the escalation of the still ongoing (although often forgotten) civil conflict.

Accordingly, both climate change and armed conflicts are major security concerns, jeopardizing human development and affecting, disproportionately, crucial aspects of human life. For this reason, during the last years a possible linkage between these two main issues has been widely debated to shed light on the possible role of climate change in inducing or aggravating armed conflicts. Hsiang and Burke (2014) review 50 independent studies examining the impacts of climate change on

conflicts and find strong support for a causal link ranging across regions, time intervals and spatial scales. However, the debate about the quantification of the role of climate change in inducing armed conflicts, as well as the mechanisms through which it operates, is still open. Scholars generally agree on the fact that climate change has an indirect effect, exacerbating already existing tensions, while the mechanisms leading to conflicts are manifold. Among these, the most recurrent in the literature include the increasing climate-induced resource scarcity, which further constraints access of vulnerable groups to water resources and productive lands and fosters competitions over their control (e.g. Maxwell and Reuveny, 2000); the so-called resource curse hypothesis, which contends that the uneven distribution of valuable resources and revenues may cause armed conflicts (Holmberg, 2008); food insecurity caused by the elevated exposure of the agricultural sector to climate change and the related diminishing crop productivity and yields (e.g. Burke et al., 2009); the social instability posed by the arrival of migrants in already fragile locations (Olzak, 1992).

Migrations

Similarly to armed conflicts, climate change is generally acknowledged to foster international and intra-country migrations, but the relative importance of its contribution compared to other causes is still hotly debated (de Sherbinin et al., 2011a). Several mechanisms owing to climate change may force people to migrate, everything else being equal. Among these are, for instance, food insecurity and the deterioration of land productivity and rural livelihoods in consequence of long-term droughts, or displacement in the aftermath of extreme events like floods, landslide, or wildfires that forces relocation and rebuilding somewhere else (Adger et al., 2014). A great source of uncertainty is also related to the quantification of migration flows under future climate change scenarios. Further, the current increase in temperatures has brought about an intensification of migratory flows mostly concentrated in some regions like Sub-Saharan Africa as well as islands and coastal areas (Brown, 2008). Nonetheless, temperature increases ranging between 2° and 4° degrees, as projected by the IPCC, may completely overturn known patterns of displacement, likely requiring different policies than the current ones (Gemenne, 2011). In addition, displacement may also be the result of the implementation of mitigation and adaptation policies, especially in the case of the creation of large projects (de Sherbini et al., 2011a).

Migration is now acknowledged by many to be a form of adaptation (McLeman and Smit, 2006; Tacoli, 2009) or as a failure to adapt (de Sherbinin et al., 2011b). However, the many other motives inducing people to migrate and the uncertainty in reconducting some indirect mechanisms fostering climate migration poses problems in the recognition of the status of climate refugees and migrants. People who resort to migration are not usually the poorest and most vulnerable households (Black et

al., 2011), since also migrating is expensive, but those who either do not have access to land (Obeng-Odoom, 2017; Kosec et al., 2018) or had access to a land that has now become unproductive.

Several studies investigate empirically climate-induced displacement focusing either on ruralurban (e.g. Findley, 1994; Henry et al., 2003; Barrios et al., 2006; Mueller and Osgood, 2009; Saldaña-Zorrilla and Sandberg, 2009; Nawrotzki et al., 2017) or international migration (e.g. Munshi, 2003; Feng et al., 2010; Marchiori and Schumacher, 2011; Maurel and Tuccio, 2016). Marchiori et al. (2012) argue that rural-urban and international migrations constitute different views of the same phenomenon and suggest looking at the overall picture. Accordingly, people initially move from rural areas, made increasingly uninhabitable in consequence of climate hazards and environmental stresses, to urban areas. Then, the increasing pressure exerted by rural migrants on urban areas, favours international migration (Marchiori et al., 2012).

6 Discussion and conclusions

With rapidly changing climate, extreme events are projected to occur with still increasing frequency and intensity in the coming years. This can seriously foster the vulnerability of poorer individuals and population groups, as well as increase the number of people endangered and fallen into poverty. In some situations, the repeated frequency of such events may not consent a full recovery and preparedness ahead of the next event, further exacerbating vulnerabilities. In this framework, strengthening local adaptive capacity, primarily facilitating access to key resources, is a mechanism through which vulnerable individuals and population groups may reinforce their resilience and be prepared for multiple events to come. To do so, it is important to recognize both climate change as a man-made process and vulnerabilities as a social construction. The result is a complex twist of interrelations between nature and society that, driven by power relations that shape the degree of social and economic inequalities, and further complicated by the additional interweaving among climate-induced impacts on wellbeing dimensions, determines winners and losers. Diverse typologies of vertical and horizontal inequalities interact with one another creating multi-faceted vulnerabilities to the impacts of climate change. For instance, women in the upper castes in India rarely lack access to relevant resources and suffer damages from extreme events. On the contrary, access to resources for women in the lower castes are doubly constrained by their caste belonging and male-female relationships which limit their freedom to act and react (Ray-Bennett, 2009). Further, climate change impacts on one dimension of wellbeing may likely cause other dimensions to be affected, as highlighted in Section 2.5.

It is therefore unthinkable to solve the climate crisis without addressing inequalities and vulnerabilities because the benefits obtained by acting on only one of these two aspects risks being

neutralized by not acting on the other. Accordingly, undertaking a sustainable transition while neglecting issues related to climate justice can bear skewed outcomes in terms of vulnerability: some may gain from such transition, while others may result increasingly vulnerable. An example is the cobalt and coltan curse that has affected Democratic Republic of Congo (DRC) in the last decades. The country hosts some of the major mines of these two metals, which are essential for the production of mobiles and electric vehicles batteries. As a result, the increasing diffusion in developed countries of electric vehicles, an undoubtedly cleaner alternative to conventional automobiles, has led to numerous conflicts to take control of these resources in DRC (Lalji, 2007). Policy responses should therefore be downscaled at the local level, where vulnerability is experienced and should importantly be evaluated systemically, considering the crossroads among different socioeconomic impacts and population groups.

Further, ignoring inequalities and vulnerabilities can have skewed outcomes not only in terms of adaptation, but of climate change mitigation as well. As Wilkinson and Pickett (2010, p. 217) contend, "given what inequality does to a society, and particularly how it heightens competitive consumption, it looks not only as if the two are complementary, but also that governments may be unable to make big enough cuts in carbon emissions without also reducing inequality". Jorgenson et al. (2017) demonstrate that higher income and wealth inequality in the USA is correlated with higher average CO₂ emissions. Jorgenson (2015) shows that the Veblen effect on consumption induced by income inequality (Bowles and Park, 2005), according to which households in the lower tails of the distribution emulate consumption of wealthier households, in turn results in increased energy use and emissions. These results are confirmed and extended globally by the very recent report elaborated by Oxfam and the Stockholm Environment Institute (Gore, 2020), according to which the richest 10% of the world population is responsible for 52% of GHG emitted in the atmosphere between 1990 and 2015. With this in mind, policies to mitigate GHG emissions should account for the differentiated effects they would have on households in the different quantiles of income distribution. To this aim, before implementing a carbon tax based on carbon content of goods and services, which would probably regressively affect poorer households more than wealthier ones, policy makers may evaluate the effects that a tax on luxury goods would have on the reduction of both emissions and inequality.

7 References

- Adger, W.N., & Kelly, P.M. (1999). Social vulnerability to climate change and the architecture of entitlements. Mitigation and adaptation strategies for global change, 4(3-4), 253-266.
- Adger, W.N., Pulhin, J.M., Barnett, J., Dabelko, G.D., Hoversrud, G.K., Levy, M., Oswald Spring, Ú. & Vogel, C.H. (2014). Human security. In: Field, C.B., Barros, V.R., Dokken, D.J., et al. (eds) Climate change 2014: impacts, adaptation, and vulnerability. Part A: Global and sectoral aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge.
- Ahmad, J., Alam, D., & Haseen, M. S. (2011). Impact of climate change on agriculture and food security in India. International Journal of Agriculture, Environment and Biotechnology, 4(2), 129-137.
- Amarasinghe U.A., Samad M., Anputhas M. (2005). Locating the poor: spatially disaggregated poverty maps for Sri Lanka. Report No. 96, International Water Management Institute, Colombo.
- Al Nahian, M., Islam, G. M. T., & Bala, S. K. (2013). A new approach in gender vulnerability assessment using matrix framework. In: Proceedings of the 4th International Conference on Water & Flood Management (ICWFM-2013). Dhaka, India: Institute of Water and Flood Management.
- Arora-Jonsson, S. (2011). Virtue and vulnerability: Discourses on women, gender and climate change. Global environmental change, 21(2), 744-751.
- Ash, M., & Boyce, J. K. (2018). Racial disparities in pollution exposure and employment at US industrial facilities. Proceedings of the National Academy of Sciences, 115(42), 10636-10641.
- Bankoff, G., Frerks, G., & Hilhorst, D. (Eds.). (2004). Mapping vulnerability: Disasters, development and people. London: Earthscan.
- Banzhaf, S., Ma, L., & Timmins, C. (2019). Environmental justice: The economics of race, place, and pollution. Journal of Economic Perspectives, 33(1), 185-208.
- Barnett, J. & O'Neill, S. (2010). Maladaptation. Global environmental change, 20, 211-213.
- Barrett, S. (2013). Local level climate justice? Adaptation finance and vulnerability reduction. Global Environmental Change, 23(6), 1819-1829.
- Barrios, S., Bertinelli, L., & Strobl, E. (2006). Climatic change and rural-urban migration: The case of sub-Saharan Africa. CORE Discussion Paper No. 2006/46.
- Bartlett, S. (2008). Climate change and urban children: impacts and implications for adaptation in low- and middle-income countries. Environment and Urbanization, 20(2), 501-519.
- Birkmann, J. (2007). Measuring vulnerability to natural hazards: towards disaster resilient societies. United Nations University Press. Tokyo, Japan.

- Birthal, P. S., Khan, T., Negi, D. S., & Agarwal, S. (2014). Impact of climate change on yields of major food crops in India: Implications for food security. Agricultural Economics Research Review, 27(2), 145-155.
- Black, R., Kniveton, D., & Schmidt-Verkerk, K. (2011). Migration and climate change: towards an integrated assessment of sensitivity. Environment and Planning A, 43(2), 431-450.
- Bohle, H.G., Downing, T.E., & Watts, M.J. (1994). Climate change and social vulnerability: toward a sociology and geography of food insecurity. Global environmental change, 4(1), 37-48.
- Boko, M., Niang, I., Nyong, A., Vogel, A., Githeko, A., Medany, M., ... & Yanda, P. Z. (2018). Africa Climate Change 2007: Impacts, Adaptation and Vulnerability: Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change.
- Bonilla-Silva, E. (2013). Racism without racists: Color-blind racism and the persistence of racial inequality in America. Lanham, MD: Rowman & Littlefield.
- Bowles, S., & Park, Y. (2005). Emulation, inequality, and work hours: Was Thorsten Veblen right?. The Economic Journal, 115(507), F397-F412.
- Brown, O. (2008). Migration and climate change (No. 31). United Nations Publications.
- Buhaug, H. (2010). Climate not to blame for African civil wars. Proceedings of the National Academy of Sciences of the United States of America, 107(38): 16477–16482.
- Bullard, R. D., & Wright, B. (Eds.). (2009). Race, place, and environmental justice after Hurricane Katrina:
 Struggles to reclaim, rebuild, and revitalize New Orleans and the Gulf Coast (pp. xix–xxii).
 Philadelphia, PA: Westview Press.
- Burke, M. B., Miguel, E., Satyanath, S., Dykema, J. A., Lobell, D. B. (2009). Warming Increases the Risk of Civil War in Africa. Proceedings of the National Academy of Sciences of the United States of America, 106(49): 20670-20674.
- Cannon, T. (1994). Vulnerability analysis and the explanation of natural disasters. In Varley, A. (ed.), Disasters Development and Environment, John Wiley, Chichester pp. 13-30.
- Cannon, T. (2006). Vulnerability analysis, livelihoods and disasters. In: Ammann, W.J., S. Dannenmann, and L. Vulliet (eds.). Risk 21: Coping with Risks Due to Natural Hazards in the 21st Century. Taylor and Francis Group, London, UK, pp. 41-49.
- Cardona, O.D., Hurtado, J.E. Duque, G. Moreno, A. Chardon, A.C. Velásquez, L.S. and Prieto, S.D. (2003). The Notion of Disaster Risk: Conceptual Framework for Integrated Management. National University of Colombia / Inter-American Development Bank, Washington, DC.
- Cardona, O.D., van Aalst, M.K., Birkmann, J., Fordham, M., McGregor, G., Perez, R., Pulwarty, R.S., Schipper, E.L.F. & Sinh, B.T. (2012). Determinants of risk: exposure and vulnerability. In: Field, C.B., V. Barros, T.F. Stocker, D. Qin, D.J. Dokken, K.L. Ebi, M.D. Mastrandrea, K.J. Mach, G.-K.

Plattner, S.K. Allen, M. Tignor, and P.M. Midgley (eds.). Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change (IPCC). Cambridge University Press, Cambridge, UK, and New York, NY, USA.

- Carr, E. R. (2008). Between structure and agency: Livelihoods and adaptation in Ghana's Central Region. Global Environmental Change, 18(4), 689-699.
- Challinor, A., Wheeler, T., Garforth, C., Craufurd, P., & Kassam, A. (2007). Assessing the vulnerability of food crop systems in Africa to climate change. Climatic change, 83(3), 381-399.
- Cinner, J. E., Adger, W. N., Allison, E. H., Barnes, M. L., Brown, K., Cohen, P. J., ... & Marshall, N. A. (2018). Building adaptive capacity to climate change in tropical coastal communities. Nature Climate Change, 8(2), 117-123.
- Cohen, M. J., Pinstrup-Andersen, P. (1999). Food security and conflict. Social Research, 375-416.
- Collier, P. (1999), On the economic consequences of civil war, Oxford Economic Papers, 51: 168-183.
- Collier, P. (2003) Breaking the Conflict Trap: Civil War and Development Policy, World Bank Policy Research Report No. 56793, World Bank Publications: Washington DC.
- Cooper, P.J.M., Dimes, J., Rao, K.P.C., Shapiro, B., Shiferaw, B. and Twomlow, S. (2008). Coping better with current climatic variability in the rain-fed farming systems of sub-Saharan Africa: An essential first step in adapting to future climate change? Agriculture, Ecosystems and Environment, 126, 24-35.
- Costello, A., Abbas, M., Allen, A., Ball, S., Bell, S., Bellamy, R., ... & Lee, M. (2009). Managing the health effects of climate change: lancet and University College London Institute for Global Health Commission. The Lancet, 373(9676), 1693-1733.
- Cretney, R. M. (2017). Towards a critical geography of disaster recovery politics: Perspectives on crisis and hope. Geography Compass, 11(1), e12302.
- Cutter, S. L., & Finch, C. (2008). Temporal and spatial changes in social vulnerability to natural hazards. Proceedings of the National Academy of Sciences, 105(7), 2301-2306.
- D'Alisa, G., & Kallis, G. (2016). A political ecology of maladaptation: Insights from a Gramscian theory of the State. Global Environmental Change, 38, 230-242.
- Das Gupta, M. (2013). Population, poverty, and climate change. The World Bank.
- de Sherbinin, A., Castro, M., Gemenne, F., Carnea, M.M., Adamo, S., Fearnside, P.M., Krieger, G., Lahmani, S., Oliver-Smith, A., Pankhurst, A., Scudder, T., Singer, B., Tan, Y., Wannier, G., Boncour, P., Ehrhart, C., Hugo, G., Pandey, B. & Shi, G. (2011a). Preparing for resettlement associated with climate change. Science 334(6055):456–457. doi:10.1126/science.1208821

- De Sherbinin, A., Warner, K., & Ehrhart, C. (2011b). Casualties of climate change. Scientific American, 304(1), 64-71.
- Denton, F. (2002). Climate change vulnerability, impacts, and adaptation: Why does gender matter? Gender and Development, 10(2), 10–20.
- Dilley, M., Chen, R. S., Deichmann, U., Lerner-Lam, A. L. & Arnold, M. (2005). Natural disasters hot spots: a global risk analysis. World Bank. Washington, DC, USA.
- Doherty, T. J., Clayton, S. (2011). The psychological impacts of global climate change. American Psychologist, 66(4): 265.
- Driesen, D. M. et al. (2005). An unnatural disaster: The aftermath of Hurricane Katrina Center for Progressive Reform (CPR) Publication #512. Washington D.C.: CPR.
- Ebi, K. L., & Bowen, K. (2016). Extreme events as sources of health vulnerability: Drought as an example. Weather and Climate Extremes, 11, 95-102.
- Eckstein, D., Künzel, V. & Schäfer, L. (2017). Global Climate Risk Index 2018: Who suffers most from extreme weather events? Weather-related loss events in 2016 and 1997 to 2016. Germanwatch. Berlin, Germany.
- Elliott, J.R. and Pais, J. (2006). Race, class, and Hurricane Katrina: social differences in human responses to disaster. Social Science Research, 35(2), 295-321.
- Enarson, E. and B.H. Morrow, 1998: The Gendered Terrain of Disaster: Through Women's Eyes. Praeger Publishers, Westport, CT, USA, 275 pp
- Eriksen, S. and Silva, J.A. (2009). The vulnerability context of a savanna area in Mozambique: household drought strategies and responses to economic change. Environmental Science and Policy, 12, 33-52.
- Eriksen, S., & Lind, J. (2009). Adaptation as a political process: adjusting to drought and conflict in Kenya's drylands. Environmental management, 43(5), 817-835.
- Etkin, D. (1999). Risk transference and related trends: Driving forces towards more megadisasters. Environmental Hazards, 1(2), 69–75.
- Feng, S., Krueger, A. B., & Oppenheimer, M. (2010). Linkages among climate change, crop yields and Mexico–US cross-border migration. Proceedings of the National Academy of Sciences, 107(32), 14257-14262.
- Findley, S. E. (1994). Does drought increase migration? A study of migration from rural Mali during the 1983–1985 drought. International Migration Review, 28(3), 539-553.
- Food and Agriculture Organization (FAO). (1996). Rome declaration on world food security and World Food Summit plan of action, World Food Summit, Rome, 13 to 17 November 1996.
- Fordham, M. (1999). Participatory planning for flood mitigation: Models and approaches. The Australian Journal of Emergency Management, 13(4), 27–34.

- Fothergill, A., Maestas, E.G.M. and Darlington, J.D. (1999). Race, ethnicity and disasters in the United States: a review of the literature. Disasters, 23(2), 156-173.
- Fukuda-Parr, S., Ashwill, M., Chiappa, E., Messineo, C. (2011). The Conflict–Development Nexus: A Survey of Armed Conflicts in Sub-Saharan Africa, 1980–2005. In Security and Development. Edward Elgar Publishing.
- Füssel, H.M. (2004). Coevolution of the political and conceptual frameworks for climate change vulnerability assessments. In: Biermann, F., Campe, S., Jacob, K. (Eds.), Proceedings of the 2002 Berlin Conference on the Human Dimensions of Global Environmental Change "Knowledge for the Sustainability Transition. The Challenge for Social Science". Global Governance Project. Amsterdam, The Netherlands, pp. 302–320.
- Füssel, H.M. (2012). Vulnerability to climate change and poverty. In Climate Change, Justice and Sustainability (pp. 9-17). Springer, Dordrecht.
- Gaillard, J. C., Glantz, M., Kelman, I., Wisner, B., Delica-Willison, Z., & Keim, M. (2014). Taking the 'naturalness' out of natural disaster (again): What about climate change?. Natural Hazards Observer, 38(3), 1-14.
- Gaillard, J.C. (2010). Vulnerability, capacity and resilience: perspectives for climate and development policy. Journal of International Development: The Journal of the Development Studies Association, 22(2), 218-232.
- Gates, S., Hegre, H., Nyga°rd, H.M., Strand, H. (2012). Development consequences of armed conflict. World Development, 40: 1713–22.
- Gemenne, F. (2011). Climate-induced population displacements in a 4 C+ world. Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences, 369(1934), 182-195.
- Gilbert, C. (1995). Studying disaster: a review of the main conceptual tools. International Journal of Mass Emergencies and Disasters, 13(3), 231-240.
- Gleick, P. H. (2014). Water, drought, climate change, and conflict in Syria. Weather, Climate, and Society, 6(3): 331-340.
- Gore, T. (2020). Confronting Carbon Inequality: Putting climate justice at the heart of the COVID-19 recovery.
- Gosling, S.N., McGregor, G. & Lowe, J. (2009). Climate change and heat-related mortality in six cities Part
 2: climate model evaluation and projected impacts from changes in the mean and variability of temperature with climate change. International Journal of Biometeorology, 53(1), 31-51.
- Gosling, S.N. and Arnell, N.W. (2013). A global assessment of the impact of climate change on water scarcity. Climatic Change, published online 17 August 2013, doi:10.1007/s10584-013-0853-x.

- Grundy, E. (2006). Ageing and vulnerable elderly people: European perspectives. Ageing and Society, Volume 26, pp. 105-34
- Hajat, S., Kovats, R. S. & Lachowycz, K. (2007). Heatrelated and cold-related deaths in England and Wales: who is at risk?. Occupational and Environmental Medicine, 64(2), pp. 93-100.
- Hallegatte, S. (2012). An exploration of the link between development, economic growth, and natural Risk. Policy Research Working Paper 6216 Washington, DC: World Bank.
- Hardy, R. D., Milligan, R. A., & Heynen, N. (2017). Racial coastal formation: The environmental injustice of colorblind adaptation planning for sea-level rise. Geoforum, 87, 62–72. https://doi.org/10.1016/j.geoforum.2017.10.005
- Harris, P.G. (2013), What's Wrong with Climate Politics and How to Fix it. Polity Press, Cambridge, MA.
- Heijmans, A. (2001). 'Vulnerability': a matter of perception (London, Benfield Greig Hazard Research Centre).
- Henry, S., Boyle, P., & Lambin, E. F. (2003). Modelling inter-provincial migration in Burkina Faso, West Africa: the role of socio-demographic and environmental factors. Applied Geography, 23(2-3), 115-136.
- Hickel, J. (2017). The divide: A brief guide to global inequality and its solutions. Random House.
- Hoegh-Guldberg, O., Jacob, D., Bindi, M., Brown, S., Camilloni, I., Diedhiou, A., ... Hijioka, Y. (2018).
 Impacts of 1.5 C global warming on natural and human systems. Global warming of 1.5° C. An IPCC Special Report.
- Holmberg, J. (2008). Natural resources in sub-Saharan Africa: assets and vulnerabilities. Current Africa Issue 39/2008. Uppsala: Nordiska Afrikainstitutet.
- Honda, Y., Kondo, M., McGregor, G., Kim, H., Guo, Y. L., Hijioka, Y., ... & Kovats, R. S. (2014). Heatrelated mortality risk model for climate change impact projection. Environmental health and preventive medicine, 19(1), 56-63.
- Hore, K., Kelman, I., Mercer, J., & Gaillard, J. C. (2018). Climate change and disasters. In Handbook of disaster research (pp. 145-159). Springer, Cham.
- Houghton, R.M.E., (2009). Domestic violence reporting and disasters in New Zealand. Regional Development Dialogue, 30(1), 79-90.
- Hsiang, S. M., Burke, M. (2014). Climate, conflict, and social stability: what does the evidence say? Climatic Change, 123(1): 39–55.
- Hsiang, S. M., Burke, M., Miguel, E. (2013). Quantifying the influence of climate on human conflict. Science, 341(6151).
- Hsiang, S., Kopp, R., Jina, A., Rising, J., Delgado, M., Mohan, S., ... & Larsen, K. (2017). Estimating economic damage from climate change in the United States. Science, 356(6345), 1362-1369.

- Inter-American Development Bank (2006). Indicators of disaster risk and risk management: summary report. Inter-American Development Bank. Washington DC, USA.
- International Strategy for Disaster Reduction (2004). Living with risk: a global review of disaster reduction initiatives. United Nations, New York.
- IPCC. (2001). Climate Change 2001: A Synthesis Report. A Contribution of Working Groups I, II and III to the Third Assessment Report of the IPCC, Watson, R.T. (ed.) and the Core Writing Team, Cambridge University Press, New York, NY.
- IPCC (2007). Climate change 2007: the physical science basis: contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge; New York: Cambridge University Press.
- IPCC. (2012). Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change [Field, C.B., V. Barros, T.F. Stocker, D. Qin, D.J. Dokken, K.L. Ebi, M.D. Mastrandrea, K.J. Mach, G.-K. Plattner, S.K. Allen, M. Tignor, and P.M. Midgley (eds.)]. Cambridge University Press, Cambridge, UK, and New York, NY, USA, 582 pp.
- IPCC. (2014). Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Field, C.B., V.R. Barros, D.J. Dokken, K.J. Mach, M.D. Mastrandrea, T.E. Bilir, M. Chatterjee, K.L. Ebi, Y.O. Estrada, R.C. Genova, B. Girma, E.S. Kissel, A.N. Levy, S. MacCracken, P.R. Mastrandrea, and L.L.White (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, 1132 pp.
- Jabry, A. (2003). Children in Disasters: After the Cameras have Gone. Plan UK, London, UK, 56 pp.
- Jorgenson, A. K. (2015). Inequality and the carbon intensity of human well-being. Journal of Environmental Studies and Sciences, 5(3), 277-282
- Jorgenson, A., Schor, J., & Huang, X. (2017). Income inequality and carbon emissions in the United States: a state-level analysis, 1997–2012. Ecological Economics, 134, 40-48.
- Jurgilevich, A., Räsänen, A., Groundstroem, F., & Juhola, S. (2017). A systematic review of dynamics in climate risk and vulnerability assessments. Environmental Research Letters, 12(1), 013002.
- Kallis, G. (2008). Droughts. Annual review of environment and resources, 33, 85-118.
- Kallis, G., & Zografos, C. (2013). Hydro-climatic change, conflict and security. Climatic change, 123(1), 69-82.
- Kelley, C. P., Mohtadi, S., Cane, M. A., Seager, R., Kushnir, Y. (2015). Climate Change In The Fertile Crescent And Implications Of The Recent Drought In Syria. In AGU Fall Meeting Abstracts.

- Kelman, I., Gaillard, J. C., & Mercer, J. (2015). Climate change's role in disaster risk reduction's future:Beyond vulnerability and resilience. International Journal of Disaster Risk Science, 6(1), 21-27.
- Khan, A. E., Xun, W. W., Ahsan, H., & Vineis, P. (2011). Climate change, sea-level rise, & health impacts in Bangladesh. Environment: Science and Policy for Sustainable Development, 53(5), 18-33.
- King, M. D. (2015). The weaponization of water in Syria and Iraq. The Washington Quarterly, 38(4): 153-169.
- Kosec, K., Ghebru, H., Holtemeyer, B., Mueller, V., & Schmidt, E. (2018). The effect of land access on youth employment and migration decisions: Evidence from rural Ethiopia. American Journal of Agricultural Economics, 100(3), 931-954.
- Kotir, J. H. (2011). Climate change and variability in Sub-Saharan Africa: a review of current and future trends and impacts on agriculture and food security. Environment, Development and Sustainability, 13(3), 587-605.
- Kunz, M. B., Mühr, T., Kunz-Plapp, J. E., Daniell, J. E., Khazai, B., Wenzel, F., ... Zschau, J. (2013). Investigation of Superstorm Sandy 2012 in a multi-disciplinary approach. Natural Hazards and Earth System Sciences, 13, 2579–2598.
- Lalji, N. (2007). The resource curse revised: Conflict and Coltan in the Congo. Harvard International Review, 29(3), 34.
- Laurent, E. (2011). Issues in environmental justice within the European Union. Ecological Economics, 70(11), 1846-1853.
- Lavell, A. (2003). Local Level Risk Management: Concept and Practices. CEPREDENAC-UNDP, Quito, Ecuador.
- Lewis, J. (1999). Development in disaster-prone places: Studies of vulnerability. London: Intermediate Technology Publications.
- Lobell, D. B., Schlenker, W., & Costa-Roberts, J. (2011). Climate trends and global crop production since 1980. Science, 333(6042), 616-620.
- Marchiori, L., & Schumacher, I. (2011). When nature rebels: international migration, climate change, and inequality. Journal of Population Economics, 24(2), 569-600.
- Marchiori, L., Maystadt, J. F., & Schumacher, I. (2012). The impact of weather anomalies on migration in sub-Saharan Africa. Journal of Environmental Economics and Management, 63(3), 355-374.
- Markandya, A., Costantini, V., Paglialunga, E., Sforna, G. (2017). "Global and Regional Economic Damages from Climate Change", Oxford Encyclopaedia of Environmental Science. OUP: New York. DOI: 10.1093/acrefore/9780199389414.013.409
- Masozera, M., Bailey, M., & Kerchner, C. (2007). Distribution of impacts of natural disasters across income groups: A case study of New Orleans. Ecological Economics, 63, 299–306.

- Maurel, M., & Tuccio, M. (2016). Climate instability, urbanisation and international migration. The Journal of Development Studies, 52(5), 735-752.
- Maxwell, J. W., Reuveny, R. (2000). Resource scarcity and conflict in developing countries. Journal of Peace Research, 37(3): 301-322.
- McCarthy, J.J., Canziani, O.F., Leary, N.A., Dokken, D.J., & White, K.S. (2001). Climate change 2001: impacts, adaptation, and vulnerability: contribution of Working Group II to the third assessment report of the Intergovernmental Panel on Climate Change (Vol. 2). Cambridge University Press.
- McKenzie, L. M., Allshouse, W. B., Burke, T., Blair, B. D., & Adgate, J. L. (2016). Population size, growth, and environmental justice near oil and gas wells in Colorado. Environmental Science & Technology, 50(21), 11471-11480.
- McLeman, R., Smit, B. (2006). Migration as an adaptation to climate change. Climatic change, 76(1-2): 31-53.
- Mearns, R., & Norton, A. (2009). Social dimensions of climate change: equity and vulnerability in a warming world. World Bank: Washington, DC.
- Messer, E., Cohen, M. J. (2004). Breaking the links between conflict and hunger in Africa (No. 26). International Food Policy Research Institute (IFPRI).
- Mueller, V. A., & Osgood, D. E. (2009). Long-term consequences of short-term precipitation shocks: evidence from Brazilian migrant households. Agricultural Economics, 40(5), 573-586.
- Munshi, K. (2003). Networks in the modern economy: Mexican migrants in the US labor market. The Quarterly Journal of Economics, 118(2), 549-599.
- Mustafa, D., Ahmed, S., Saroch, E., & Bell, H. (2010). Pinning down vulnerability: From narratives to numbers. Disasters, 35(1), 62–86.
- Nagoda, S., & Nightingale, A. J. (2017). Participation and power in climate change adaptation policies: Vulnerability in food security programs in Nepal. World Development, 100, 85-93.
- Nawrotzki, R. J., DeWaard, J., Bakhtsiyarava, M., & Ha, J. T. (2017). Climate shocks and rural-urban migration in Mexico: exploring nonlinearities and thresholds. Climatic change, 140(2), 243-258.
- Nelson G.C., Palazzo A., Ringler C., Sulser T., Batka M. (2009). The role of international trade in climate change adaptation. International Centre For Trade And Sustainable Development (ICTSD) and the International Food & Agricultural Trade Policy Council (IPC), Geneva/Washington.
- Newell, B., Crumley, C. L., Hassan, N., Lambin, E. F., Pahl-Wostl, C., Underdal, A., & Wasson, R. (2005). A conceptual template for integrative human–environment research. Global Environmental Change, 15(4), 299-307.
- Obeng-Odoom, F. (2017). Unequal access to land and the current migration crisis. Land Use Policy, 62, 159-171.

- OECD (Organisation for Economic Co-operation and Development) (2015). States of Fragility 2015: Meeting Post-2015 Ambitions. Paris: OECD.
- Olzak, S. (1992). The Dynamics of Ethnic Competition and Conflict (Stanford University Press). Stanford.
- Onta, N., & Resurreccion, B. P. (2011). The role of gender and caste in climate adaptation strategies in Nepal. Mountain Research and Development, 31(4), 351-356.
- Oppenheimer, M., Campos, M., Warren, R., Birkmann, J., Luber, G., O'Neill, B., ... & Hsiang, S. (2014). Emergent risks and key vulnerabilities. In Climate Change 2014 Impacts, Adaptation and Vulnerability: Part A: Global and Sectoral Aspects (pp. 1039-1100). Cambridge University Press.
- Otto, I. M., Reckien, D., Reyer, C. P., Marcus, R., Le Masson, V., Jones, L., ... & Serdeczny, O. (2017). Social vulnerability to climate change: a review of concepts and evidence. Regional Environmental Change, 17: 1651–1662.
- Peacock,W., (1997). HurricaneAndrew: Ethnicity, Gender and the Sociology of Disasters. Routledge, Abingdon, UK and New York, NY, USA, 304 pp.
- Pfeffer, N., Wen, F. H., Ikhrata, H. M., & Gosnell, J. R. (2002). Environmental justice in the transportation planning process: Southern California perspective. Transportation Research Record, 1792(1), 36-43.
- Pradhan, E.K., West, K., Katz, J., LeClerq, S.C., Khatry, S.K. & Shrestha, S.R. (2007). Risk of flood-related mortality in Nepal. Disasters 31(1):57–70.
- Ray-Bennett, N. S. (2009). The influence of caste, class and gender in surviving multiple disasters: A case study from Orissa, India. Environmental Hazard, 8(1), 5–22. https://doi.org/10.3763/ehaz.2009.0001
- Saldaña-Zorrilla, S. O., & Sandberg, K. (2009). Impact of climate-related disasters on human migration in Mexico: a spatial model. Climatic change, 96(1-2), 97.
- Scandrett, E. (2016). Climate justice: contested discourse and social transformation. International Journal of Climate Change Strategies and Management.
- Schewe, J., Heinke, J. Gerten, D. Haddeland, I. Arnell, N.W. Clark, D.B. Dankers, R. Eisner, S. Fekete, B.M.
 Colón-González, F.J. Gosling, S.N. Kim, H. Liu, X. Masaki, Y. Portmann, F.T. Satoh, Y. Stacke, Y.
 Tang, Q. Wada, Y. Wisser, D. Albrecht, T. Frieler, K. Piontek, F. Warszawski, L. & Kabat, P. (2013).
 Multi-model assessment of water scarcity under climate change. Proceedings of the National
 Academy of Sciences of the United States of America. Published ahead of print 16 December 2013, doi:10.1073/pnas.1222460110.
- Schmidt, A. (Ed.). (2005). Linking poverty reduction and disaster risk management.
- Schröter, D., Polsky, C. & Patt, A. (2005). Assessing vulnerabilities to the effects of global change: an eight step approach. Mitigation and Adaptation Strategies for Global Change 10 (4), 573–595.
- Seed. R. (2007). Re: New Orleans, Hurricane Katrina, and the soul of the profession [Letter from Ray Seed, Professor of Civil and Environmental Engineering to William Marcuson III, President, American

SocietyofCivilEngineers].Availableat:http://aquadoc.typepad.com/files/katrina_and_soul_of_the_profession.pdf.

- Sen, A. K. (1981). Poverty and famines: an essay on entitlement and deprivation. Oxford University Press.
- Sen, A. K. (1987). Property and hunger. Economics and philosophy, vol. 4, pp. 57-68.
- Sen, A. K. (1990). Food, economics and entitlements. In Dreze, J. and Sen, A. K. (eds.), The political economy of hunger, vol. 1, pp. 34-50.
- Sena, A., Barcellos, C., Freitas, C., & Corvalan, C. (2014). Managing the health impacts of drought in Brazil. International journal of environmental research and public health, 11(10), 10737-10751.
- Shearer, C. (2012). The social construction of Alaska native vulnerability to climate change. Race, Gender and Class, 19(1–2), 61–79.
- Shemyakina, O. (2011). The effect of armed conflict on accumulation of schooling: Results from Tajikistan. Journal of Development Economics, 95(2): 186-200.
- Shepherd A., Mitchell T., Lewis K.K., Lenhardt A., Jones L., Scott L., Muir-Wood R. (2013). The geography of poverty, disasters and climate extremes in 2030. Overseas Development Institute, London.
- Simelton, E., Fraser, E. D., Termansen, M., Benton, T. G., Gosling, S. N., South, A., ... & Forster, P. M. (2012). The socioeconomics of food crop production and climate change vulnerability: a global scale quantitative analysis of how grain crops are sensitive to drought. Food Security, 4(2), 163-179.
- Smith, K.R., Woodward, A., Campbell-Lendrum, D., Chadee, D.D., Honda, Y., Liu, Q., Olwoch, J.M., Revich, B. & Sauerborn, R. (2014). Human health: impacts, adaptation, and co-benefits. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Field, C.B., V.R. Barros, D.J. Dokken, K.J. Mach, M.D. Mastrandrea, T.E. Bilir, M. Chatterjee, K.L. Ebi, Y.O. Estrada, R.C. Genova, B. Girma, E.S. Kissel, A.N. Levy, S. MacCracken, P.R. Mastrandrea, and L.L.White (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, pp. 709-754.
- Sovacool, B. K. (2017). Don't let disaster recovery perpetuate injustice. Nature, 549, 433.
- Sovacool, B. K. (2018). Bamboo beating bandits: Conflict, inequality, and vulnerability in the political ecology of climate change adaptation in Bangladesh. World Development, 102, 183-194.
- Sovacool, B. K., Tan-Mullins, M., & Abrahamse, W. (2018). Bloated bodies and broken bricks: Power, ecology, and inequality in the political economy of natural disaster recovery. World Development, 110, 243-255.
- Stafoggia, M., Forastiere, F., Agostini, D., Biggeri, A. Bisanti, L., Cadum, E., Caranci, N., de'Donato, F., De Lisio, S., De Maria, M., Michelozzi, P., Miglio, R., Pandolfi, P., Picciotto, S., Rognoni, M., Russo,

A., Scarnato, C. & Perucci, C.A. (2006). Vulnerability to heat-related mortality: a multicity, population-based,case-crossover analysis. Epidemiology, 17(3), 315-323

- Stanke, C., Kerac, M., Prudhomme, C., Medlock, J., & Murray, V. (2013). Health effects of drought: a systematic review of the evidence. PLoS currents, 5.
- Stewart, F. (2009). Horizontal inequality: two types of trap. Journal of Human Development and Capabilities, 10(3), 315-340.
- Stewart, F., Brown, G. K., & Mancini, L. (2005). Why horizontal inequalities matter: some implications for measurement. Centre for Research on Inequality, Human Security and Ethnicity.
- Sultana, F. (2010). Living in hazardous waterscapes: gendered vulnerabilities and experiences of floods and disasters. Environmental Hazards, 9(1), 43-53.
- Sultana, F. (2014). Gendering climate change: Geographical insights. The Professional Geographer, 66(3), 372–381. https://doi.org/10.1080/00330124.2013.821730
- Tacoli, C. (2009). Crisis or adaptation? Migration and climate change in a context of high mobility. Environment and Urbanization, 21(2): 513–525.
- Tegat, W. J. M., Sheldon, G. W. & Griffiths, D. C. (eds.) (1990). Climate change: the IPCC impact assessment. Report prepared for Intergovernmental Panel on Climate Change by Working Group II, Australian Government Publishing, Camberra.
- Teodosijevic, S. B. (2003). Armed conflicts and food security. ESA Working Paper No. 03-11. Food and Agriculture Organization.
- Thomas, K., Hardy, R. D., Lazrus, H., Mendez, M., Orlove, B., Rivera-Collazo, I., ... & Winthrop, R. (2018). Explaining differential vulnerability to climate change: A social science review. Wiley Interdisciplinary Reviews: Climate Change, 10(2), e565.
- United Nations (2020). A new era of conflict and violence. UN75 2020 and Beyond Issue Brief.
- United Nations World Bank (2018). Pathways for Peace: Inclusive Approaches to Preventing Violent Conflict. Executive Summary booklet. World Bank, Washington, DC.
- Walters, V., & Gaillard, J. C. (2014). Disaster risk at the margins: Homelessness, vulnerability and hazards. Habitat International, 44, 211-219.
- Watts, M. J. & Bohle, H. G. (1993). The space of vulnerability: the causal structure of hunger and famine. Progress in Human Geography 17, 43-67.
- Weber, L., & Messias, D. K. H. (2012). Mississippi front-line recovery work after Hurricane Katrina: An analysis of the intersections of gender, race, and class in advocacy, power relations, and health. Social Science & Medicine, 74(11), 1833–1841.
- Weir, T., Virani, Z. (2011). Three linked risks for development in the Pacific Islands: Climate change, disasters and conflict. Climate and Development, 3(3): 193–208.

- Wheeler, T., & Von Braun, J. (2013). Climate change impacts on global food security. Science, 341(6145), 508-513.
- Wilkinson, R., & Pickett, K. (2010). The spirit level. Why equality is better for everyone.
- Wisner, B., Blaikie, P., Cannon, T. & Davis, I. (2004). At Risk, Natural Hazards, People's Vulnerability and Disasters. Routledge, London, UK.
- World Bank. (2010). World Development Report 2010: Development and Climate Change. The International Bank for Reconstruction and Development / The World Bank, Washington, DC, USA, 417 pp.
- Yamamura, E. (2015). The impact of natural disasters on income inequality: analysis using panel data during the period 1970 to 2004. International Economic Journal, 29(3), 359-374.
- Yusa, A., Berry, P., J Cheng, J., Ogden, N., Bonsal, B., Stewart, R., & Waldick, R. (2015). Climate change, drought and human health in Canada. International journal of environmental research and public health, 12(7), 8359-8412.
- Zamani, G. H., Gorgievski-Duijvesteijn, M. J., & Zarafshani, K. (2006). Coping with drought: Towards a multilevel understanding based on conservation of resources theory. Human Ecology, 34(5), 677-692.

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI Fondazione Eni Enrico Mattei Working Paper Series

Our Working Papers are available on the Internet at the following addresses: http://www.feem.it/getpage.aspx?id=73&sez=Publications&padre=20&tab=1

NOTE DI LAVORO PUBLISHED IN 2020

- 1. 2020, FACTS Series, Alessandra Celani de Macedo, Nicola Cantore, Laura Barbier, Marco Matteini, Giorgia Pasqualetto, <u>The Impact of Industrial Energy Efficiency on Economic and Social Indicators</u>
- 2. 2020, 2030 Agenda Series, Nicola Comincioli, Sergio Vergalli, <u>Effects of Carbon Tax on Electricity Price</u> <u>Volatility: Empirical Evidences from the Australian Market</u>
- 3. 2020, 2030 Agenda Series, Marco Buso, Cesare Dosi, Michele Moretto, <u>Do Exit Options Increase the</u> <u>Value-For-Money of Public-Private Partnerships?</u>
- 4. 2020, FACTS Series, Ricardo Nieva, <u>A Tragic Solution to the Collective Action Problem: Implications for</u> <u>Corruption, Conflict and Inequality</u>
- 5. 2020, FACTS Series, Charles Fang Chin Cheng, Nicola Cantore, <u>The Inclusive and Sustainable</u> <u>Development Index: a Data Envelopment Analysis Approach</u>
- 2020, 2030 Agenda Series, FACTS Series, Coker Eric, Cavalli Laura, Fabrizi Enrico, Guastella Gianni, Lippo Enrico, Parisi Maria Laura, Pontarollo Nicola, Rizzati Massimiliano, Varacca Alessandro, Vergalli Sergio, <u>The</u> <u>Effects of Air Pollution on COVID-19 Related Mortality in Northern Italy</u>
- 7. 2020, 2030 Agenda Series, Laura Cavalli, Giulia Lizzi, <u>Port of the future : Addressing Efficiency and</u> <u>Sustainability at the Port of Livorno with 5G</u>
- 8. 2020, FACTS Series, Federica Cappelli, Gianni Guastella, Stefano Pareglio, <u>Institutional Fragmentation and</u> <u>Urbanisation in the EU Cities</u>
- 2020, FEP Series, Giacomo Falchetta, Nicolò Stevanato, Magda Moner-Girona, Davide Mazzoni, Emanuela Colombo and Manfred Hafner, <u>M-LED: Multi-sectoral Latent Electricity Demand Assessment for Energy</u> <u>Access Planning</u>
- 10. 2020, Local Projects Series, Marcella De Filippo, Annalisa Percoco, Angela Voce, <u>Covid-19 e didattica a</u> <u>distanza.ll caso Basilicata, una regione a rischio digital divide</u>
- 11. 2020, 2030 Agenda, Laura Cavalli, Sandro Sanna, Mia Alibegovic, Filippo Arras, Gianluca Cocco, Luca Farnia, Emanuela Manca, Luisa F. Mulas, Marco Onnis, Sandro Ortu, Ilenia G. Romani, Marta Testa, <u>The</u> <u>Contribution of the European Cohesion Policy to the 2030 Agenda: an Application to the Autonomous</u> <u>Region of Sardinia</u>
- 12. 2020, FACTS Series, Alexander Golub, Kristina Govorukha, Philip Mayer, Dirk Rübbelke, <u>How does Climate</u> <u>Change Affect the Transition of Power Systems: the Case of Germany</u>
- 13. 2020, FEP Series, Rossana Scita, Pier Paolo Raimondi and Michel Noussan, <u>Green Hydrogen: the Holy</u> <u>Grail of Decarbonisation? An Analysis of the Technical and Geopolitical Implications of the Future</u> <u>Hydrogen Economy</u>
- 14. 2020, FACTS Series, Marta Montinaro, Rupayan Pal, Marcella Scimitore, <u>Per Unit and Ad Valorem</u> <u>Royalties in a Patent Licensing Game</u>
- 15. 2020, 2030 Agenda, S. Quaini, S. Saccani, S. Vergalli, L. Assom, M. Beria, A. Codello, M. Monaco, R. Sabatini, <u>Seasonality Fingerprint on Global Trading of Food-commodities. A Data-mining Approach</u>
- 16. 2020, 2030 Agenda Series, Nicola Comincioli, Paolo M. Panteghini and Sergio Vergalli, <u>Debt and</u> <u>Transfer Pricing: Implications on Business Tax Policy</u>
- 17. 2020, FACTS Series, Wolfgang Buchholz, Dirk Rübbelke, <u>Overstraining International Climate Finance:</u> <u>When Conflicts of Objectives Threaten Its Success</u>
- 18. 2020, 2030 Agenda Series, Nicola Comincioli, Verena Hagspiel, Peter M. Kort, Francesco Menoncin, Raffaele Miniaci and Sergio Vergalli, <u>Mothballing in a Duopoly: Evidence from a (Shale) Oil Market</u>
- 19. 2020, 2030 Agenda Series, Chiara Castelli and Angela Parenti, <u>Commuting in Europe: An Inter-regional</u> <u>Analysis on its Determinants and Spatial Effects</u>
- 20. 2020, 2030 Agenda Series, Iwan Bos, Marco A. Marini, Collusion in Quality-Segmented Markets
- 21. 2020, FACTS Series, Federica Cappelli, <u>Investigating the Origins of Differentiated Vulnerabilities to Climate</u> <u>Change and their Effects on Wellbeing</u>

Fondazione Eni Enrico Mattei

Corso Magenta 63, Milano - Italia

Tel. +39 02.520.36934 Fax. +39.02.520.36946

E-mail: letter@feem.it www.feem.it

